Author's response to reviews

Title: Is the Strongest Causal Evidence Against Customary Parental Spanking Due to Residual Confounding? A Replication of the Straus et al. (1997) Landmark Study with Alternative Disciplinary Tactics and Psychotherapy

Authors:

Robert E Larzelere (Robert.Larzelere@okstate.edu)  
Ronald B Cox Jr (r.cox@okstate.edu)  
Gail L Smith (smithg@girlsandboystown.org)

Version: 2 Date: 10 April 2009

Author's response to reviews:

I inadvertently uploaded a slightly earlier version of the manuscript. This revision includes last-minute improvements to correspond with BMC’s submission guidelines.